Author’s response to reviews

Title: Identification of a rare SEPT9 variant in a family with autosomal dominant Charcot-Marie-Tooth disease

Authors:

Gerrit M. Grosse (grosse.gerrit@mh-hannover.de)
Christine Bauer (Christine-Bauer@gmx.co.uk)
Bruno Kopp (kopp.bruno@mh-hannover.de)
Christoph Schrader (schrader.christoph@mh-hannover.de)
Alma Osmanovic (osmanovic.alma@mh-hannover.de)

Version: 4 Date: 20 Feb 2020

Author’s response to reviews:

Dear colleagues,

the submitted variant is now accessible at ClinVar. Moreover, all data of this study are thus available in the manuscript. We accordingly added the specific information as follows:

Availability of Data and Materials

The identified variant in this study has been submitted to the ClinVar database at NCBI archives (ID: SCV001156521; https://www.ncbi.nlm.nih.gov/clinvar/). All data generated or analysed during this study are included in this published article.

We thank the editorial board and the reviewers for their valuable comments on our paper.

Woth best regards

Gerrit M. Grosse (on behalf of the authors)